Author's response to reviews

Title: Psychological distress as predictor of quality of life in men experiencing infertility: a cross-sectional survey

Authors:

Juliana LR Chachamovich (jurigol@videotron.ca)
Eduardo Chachamovich (echachamovich@videotron.ca)
Hélène Ezer (helene.ezer@videotron.ca)
Fernanda P Cordova (fpcordova@terra.com.br)
Marcelo P Fleck (mfleck.voy@terra.com.br)
Daniela R Kauth (daniela.kauth@gmail.com)
Eduardo P Passos (epp@via-rs.net)

Version: 4 Date: 19 April 2010

Author's response to reviews: see over
To Regina Kuller & José Guilherme Cecatti

Editors-in-chief,

Reproductive Health

We are pleased to return the paper “Psychological distress as predictor of quality of life in men experiencing infertility: a cross-sectional survey”, following the alterations suggested by the editorial.

Please find below the alterations made in the manuscript.

Major revisions

Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. We have used the Endnote program to adjust the reference list. This program has a style, which has a style called “Reproductive Health”, and it has been used in this manuscript. In case this style has not been the appropriate style for publication, we would kindly ask to be provided by the appropriate style name.

Placement text: Please remove all figure and table placement text. All figures and tables must be cited in the manuscript, so please check that they are all still cited in order once placement text is removed. We have adjusted it in the manuscript.

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Tables included in the manuscript file need to fit on an A4 page in portrait orientation, with no visible vertical lines. We have followed this requirement and modification can be found in the manuscript.

Minor revisions: We have included the email addresses of all authors on the title page.

We are grateful for the reviewers’ and editorial’s suggestions, and we are sure that the manuscript benefited greatly from the comments.

Yours truly,

Juliana Chachamovich et al.